Bibliography

[1] J. C. Alexander, The additive inverse eigenvalue problem and topological degree, Proc. Amer. Math. Soc., 70(1978), 5-7.

[2] E. L. Allgower and K. Georg, Simplicial and continuation methods for approximating fixed points and solutions to systems of equations, SIAM Review, 22(1980), 28-85.

[3] T. W. Anderson, I. Olkin and L. G. Underhill, Generation of random orthogonal matrices, SIAM Journal on Sci. and Stat. Comput., 8(1987), 625–629.

[4] B. C. Arnold, Majorization and the Lorenz Order: A Brief Introduction, Lecture Notes in Statistics 43, Springer-Verlag, Berlin, 1987.

[5] W. W. Barrett and C. R. Johnson, Possible spectra of totally positive matrices, Linear Alg. Appl., 62(1984), 231-233.

[6] V. I. Arnold, Geometrical Methods in the Theory of Ordinary Differntial Equations, 2nd ed., Springer-Verlag, New York, 1988.

[7] A. Berman and R. J. Plemmons, Nonnegative Matrices in the Mathematical Sciences, Academic Press, New York, 1979, also Classics in Applied Mathematics 9, SIAM, Philadelphia, 1994.

[8] R. Bhatia, Perturbation Bounds for Matrix Eigenvalues, Longman Scientific & Technical, Wiley, New York, 1987.

[9] F. W. Biegler-König, A Newton iteration process for inverse eigenvalue problems, Numer. Math., 37(1981), 349-354.

[10] F. W. Biegler-König, Construction of band matrices from spectral data, Linear Alg.

Appl., 40(1981), 79-84.

[11] A. M. Bloch, R. W. Brockett and T. S Ratiu, A new formulation of the generalized Toda lattice equations and their fixed point analysis via the momentum map, Bull. Amer. Math. Soc. (N.S.) 23(1990), 477-485. 245

[12] A. M. Bloch, Steepest descent, linear programming, and Hamiltonian flows, Contemporary Math., 114(1990), 77-88.

[13] D. Boley and G. H. Golub, A modified method for reconstructing periodic Jacobi matrices, Math. Comput., 24(1984), 143-150.

[14] D. Boley and G. H. Golub, A survey of matrix inverse eigenvalue problem, Inverse Problems 3(1987), 595-622.

[15] C. de Boor and G. H. Golub, The numerically stable reconstruction of a Jacobi matrix from spectral data, Linear Alg. Appl., 21(1978), 245-260.

[16] S. Boyd, L. E. Ghaoui, E. Feron and V. Balakrishnan, Linear Matrix inequalities in System and Control Theory, 15, Studies in Applied Mathematics, SIAM, Philadelphia, 1994.

[17] J. P. Boyle and R. L. Dykstra, A method for finding projections onto the intersection of convex sets in Hilbert space, in Advances in Order Restricted Statistical Inference, Lecture Notes in Statistics, Vol. 37, ed. R. Dykstra, T. Robertston, and F. Wright, Springer-Verlag, Berlin, 1986, 28-47.

[18] M. Boyle and D. Handelman, The spectra of nonnegative matrices via symbolic dynamics, Ann. Math., 133(1991), 249-316.

[19] F. Brauer and J. A. Hohel, Qualitative Theory of Ordinary Differential Equations, Benjamin, New York, 1969.

[20] R. W. Brockett, Least squares matching problems, Linear Alg. Appl., 122/123/124(1989), 761-777.

[21] R. W. Brockett, Dynamical systems that sort lists, diagonalize matrices and solve linear programming problems, in Proc. 27th IEEE Conf. Decision and Control, (1988), 799-803, and Linear Alg. Appl. 146(91), 79-91.

[22] A. Bunse-Gerstner, R. Byers and V. Mehrmann, A chart of numerical methods for structured eigenvalue problems, Materialien LVII, Universitát Bielefeld, preprint, 1989.

[23] A. Bunse-Gerstner, R. Byers, V. Mehrmann and N. K. Nichols, Numerical computation of an analytic singular value decomposition of a matrix valued function, Numer. Math. 60(1991), 1-40.

[24] R. E. Burkard and U. Derigs, Assignment and Matching Problems: Solution Methods with FORTRAN-Programs, Springer-Verlag, Berlin Heidelberg New York 1980.

[25] R. Byers, A Hamiltonian QR algorithm, SIAM J. Sci. Stat. Comput., 7(1986), 212- 229.

[26] M. P. Calvo, A. Iserles and A. Zanna, Numerical solution of isospectral flows, DAMTP 1995/NA03, University of Cambridge, preprint.

[27] A. Cantoni and P. Butler, Eigenvalues and eigenvectors of symmetric centrosymmetric matrices, Linear Alg. Appl., 13(1976), 275-288.

[28] R. L. Causey, On Closest Normal Matrices, Ph.D. Thesis, Department of Computer Science, Stanford University, 1964.

[29] P. J. Channell and C. Scovel, Sympletic integration of Hamiltonian systems, Nonlinearity, 3(1990), 231-259.

[30] X. Chen and M. T. Chu, On the least squares solution of inverse eigenvalue problems, SIAM J. Numer. Anal., to appear.

[31] W. Cheney and A. Goldstein, Proximity maps for convex sets, Proc. Amer. Math.

Soc., 10(1959), 448-450

[32] S. N. Chow, J. Mallet-Paret and J. A. Yorke, Finding zeros of maps: Homotopy methods that are constructive with probability one, Math. Comp., 32(1978), 887- 899.

[33] M. T. Chu, The generalized Toda flow, the QR algorithm and the center manifold theorem, SIAM J. Alg. Disc. Meth., 5(1984), 187-210.

[34] M. T. Chu, A simple application of the homotopy method to symmetric eigenvalue problems, Linear Alg. Appl., 59(1984), 85-90.

[35] M. T. Chu, A continuous approximation to the generalized Schur decomposition, Linear Alg. Appl., 78(1986), 119-132.

[36] M. T. Chu, Curves on S^{n−1} that lead to eigenvalues or their means of a matrix, SIAM J. Alg. Disc. Meth., 7(1986), 425-432.

[37] M. T. Chu, On a differential equation approach to the singular value decomposition of bidiagonal matrices, Linear Alg. Appl., 80(1986), 71-79.

[38] M. T. Chu, T. Y. Li and T. Sauer, Homotopy method for general λ-matrix problems, SIAM J. Matrix Anal. Appl., 9(1988), 528-536.

[39] M. T. Chu, On the continuous realization of iterative processes, SIAM Review, 30(1988), 375-387.

[40] M. T. Chu and L. K. Norris, Isospectral flows and abstract matrix factorizations, SIAM J. Numer. Anal., 25(1988), 1383-1391.

[41] M. T. Chu, Solving additive inverse eigenvalue program for symmetric matrices by the homotopy method, IMA J. Numer. Anal., 9(1990), 331-342.

[42] M. T. Chu and K. R. Driessel, The projected gradient method for least squares matrix approximations with spectral constraints, SIAM J. Numer. Anal., 27(1990), 1050-1060.

[43] M. T. Chu, A continuous Jacobi-like approach to the simultaneous reduction of real matrices, Linear Alg. Appl., 147(1991), 75-96.

[44] M. T. Chu and K. R. Driessel, Constructing symmetric nonnegative matrices with prescribed eigenvalues by differential equations, SIAM J. Math. Anal., 22(1991), 1372-1387.

[45] M. T. Chu and M. A. Erbrecht, Symmetric Toeplitz matrices with two prescribed eigenpairs, SIAM J. Matrix Anal. Appl., 15(1994), 623-635.

[46] M. T. Chu, Numerical methods for inverse singular value problems, SIAM J. Numer.

Anal., 29(1992), 885-903.

[47] M. T. Chu, Matrix differential equations: A continuous realization process for linear algebra problems, Nonlinear Anal., TMA, 18(1992), 1125-1146.

[48] M. T. Chu and J. L. Watterson, Multivariate eigenvalue problem: I. Algebraic Theory and Power method, SIAM J. Sci. Comput., 14(1993), 1089-1106.

[49] M. T. Chu, Constructing a Hermitian matrix from its diagonal entries and eigenvalues, SIAM J. Matrix Anal. Appl., to appear, 1993.

[50] M. T. Chu, On the refinement of a Newton method for the inverse Toeplitz eigenvalue problem, SIAM J. Numer. Anal., under revision.

[51] M. T. Chu, On the Differential Equation^{dX}_{dt} = [X, k(X)] where k is a Toeplitz annihilator, J. Math. Anal. Appl., submitted.

[52] G. J. Cooper, Stability of Runge-Kutta methods for trajectory problems, IMA J.

Numer. Anal., 7(1987), 1-13.

[53] M. L. Curtis, Matrix Groups, Springer-Verlag, New York, 1979.

[54] G. Cybenko, On the eigenstructure of Toeplitz matrices, IEEE Trans. Acoust., Speech, Signal Process., ASSP-32(1984), 918-920.

[55] P. Deift, T. Nanda,and C. Tomei, Differential equations for symmetric eigenvalue eigenvalue problems, SIAM J. Numer. Anal., 20(1983), 1-22.

[56] P. Deift, J. Demmel, L. C. Li and C. Tomei, The bidiagonal singular value decomposition and the Hamiltonian mechanics, LAPACK Working Note #11, Argonne National Laboratory, 1989.

[57] J. Della-Dora, Numerical linear algorithms and group theory, Linear Alg. Appl., 10(1975),267-283.

[58] P. Delsarte and Y. Genin, Spectral properties of finite Toeplitz matrices, Proceedings of the 1983 International Symposium of Mathematical Theory of Networks and Systems, Beer-Sheva, Israel, Lecture Notes in Control and Information Sciences, 58(1984), 194-213.

[59] J. W. Demmel and A. Edelman, The dimension of matrices (matrix pencils) with given Jordan (Kronecker) canonical forms, Linear Alg. Appl., 230(1995), 47-60.

[60] J. E. Dennis, Jr. and R. B. Schnabel, Numerical Methods for Unconstrained Optimization and Nonlinear Equations, Prentice-Hall, Inc., Englewood Cliffs, 1983.

[61] F. Deutsch, Von Neumann’s alternating method: the rate of convergence, Approximation Theory IV, C. Chui, L. Schumaker and J. Ward, ed., Academic Press, New York, 1983, 427-434.

[62] L. Dieci, R. D. Russel and E. S. Van Vleck, Unitary integrators and applications to continuous orthonormalisation techniques, SIAM J. Numer. Anal., 31(1994), 261- 281.

[63] A. C. Downing and A. S. Householder, Some inverse characteristic value problems, J. Assoc. Comput. Mach., 3(1956), 203-207.

[64] T. Eirola and O. Nevanlinna, What do multistep methods approximate? Numer.

Math., 53(1988), 559-569.

[65] T. Eirola and J. M. Sanz-Serna, Conservation of integrals and symplectic structure in the integration of differential equations by multistep method, Numer. Math., 61(1992), 281-290.

[66] L. E. Faibusovich, QR-type factorizations, the Yang-Baxter equations, and an eigenvalue problem of control theory, Linear Alg. Appl., 122/123/124(1989), 943-971.

[67] L. Faybusovich, Hamiltonian structure of dynamical systems which solve linear programming problems, Physics D, 53(1991), 217-232.

[68] L. Faybusovich, Dynamical systems which solve optimization problems with linear constraints, IMA J. Math. Control Inform., 8(1991), 135-149.

[69] M. Fiedler, Eigenvalues of nonnegative symmetric matrices, Linear Algebra and Appl., 9(1974), 119-142.

[70] W. Ferguson, The construction of Jacobi and periodic Jacobi matrices with prescribed spectra, Math. Comput., 35(1980), 79-84.

[71] G. E. Forsythe and E. G. Strauss, On best conditioned matrices, Proc. Amer. Math.

Soc., 6(1955), 340-345.

[72] S. Friedland, On inverse multiplicative eigenvalue problems for matrices, Linear Alg.

Appl., 12(1975), 127-137.

[73] S. Friedland, Inverse eigenvalue problems, Linear Alg. Appl., 17(1977), 15-71.

[74] S. Friedland, On an inverse problem for nonnegative and eventually nonnegative matrices, Israel J. Math., 29(1978), 43-60.

[75] S. Friedland, The reconstruction of a symmetric matrix from the spectra data, J.

Math. Anal. Appl., 71(1979), 412-422.

[76] S. Friedland and A. A. Melkman, On the eigenvalues of nonnegative Jacobi matrices, Linear Algebra and Appl., 25(1979), 239-254.

[77] S. Friedland, J. Nocedal and M. L. Overton, The formulation and analysis of numerical methods for inverse eigenvalue problems, SIAM J. Numer. Anal., 24(1987), 634-667.

[78] S. Friedland, Inverse eigenvalue problems for symmetric Toeplitz matrices, SIAM J.

Matrix Anal. Appl., 13(1992), 1142-1153.

[79] R. Gabriel, Matrizen mit maximaler Diagonale bei unitärer Similarität, J. Reine Angew. Math., 307/308(1979), 31-52.

[80] R. Gabriel, The normal ΔH-matrices with connection to some Jacobi-like methods, Linear Alg. Appl., 91(1987), 181-194.

[81] F. R. Gantmacher, The Theory of Matrices, Vols. 1-2, Chelsea, New York, 1959.

[82] C. B. Garcia and T. Y. Li, On the number of solutions to polynomial systems of equations, SIAM J. Numer. Anal., 17(1980), 540-546.

[83] C. W. Gear, Maintaining solution invariants in the numerical solution of ODEs, SIAM J. Sci. Stat. Comput., 7(1986), 734-743.

[84] C. W. Gear, Invariants and methods for ODEs, NEC Research Institute, preprint, 1991.

[85] I. M. Gelfand and V. A. Ponomarev, Remarks on the classification of a pair of commuting linear transformations in a finite dimensional space, Functional Anal. Appl., 3(1969), 325-326.

[86] P. E. Gill, W. Murray and M. H. Wright, Practical Optimization, Academic Press, London, 1981.

[87] G. M. L. Gladwell, The inverse problem for the vibrating beam, Proc. R. Soc., A 393(1984), 277-295.

[88] G. M. L. Gladwell and J. A. Gbadeyan, On the inverse problem of the vibrating string or rod, Q. J. Mech. Appl. Math., 38(1985), 169-174.

[89] G. M. L. Gladwell, The inverse mode problem for lumped-mass systems, Q. J. Mech.

Appl. Math., 39(1986), 297-307.

[90] G. M. L. Gladwell, Inverse Problems in Vibration, Martinus Nijhoff, Dordrecht, Netherlands, 1986.

[91] W. K. Glunt, An alternating projections method for certain linear problems in a Hilbert space, preprint, University of Kentucky, 1992.

[92] A. Greenbaum and G. H. Rodrigue, Optimal preconditioners of a given sparsity pattern, BIT, 29(1989), 610-634.

[93] A. Greenbaum, Diagonally scalings of the Laplacian as preconditioners for other elliptic differential operators, SIAM J. Matrix Anal. Appl., 13(1992), 826-846.

[94] G. H. Golub, Some modified matrix eigenvalue problems, SIAM Review, 15(1973), 318-344.

[95] G. H. Golub, and C. F. Van Loan, Matrix Computations, 2nd ed., Johns Hopkins University Press, Baltimore, 1989.

[96] A. Greenbaum and G. H. Rodrigue, Optimal preconditioners of a given sparsity pattern, BIT, 29(1989), 610-634.

[97] A. Greenbaum, Diagonally scalings of the Laplacian as preconditioners for other elliptic differential operators, SIAM J. Matrix Anal. Appl., 13(1992), 826-846.

[98] R. Grone, C. R. Johnson, E. M. Sa and H. Wolkowicz, Normal matrices, Linear Alg.

Appl., 87(1987), 213-225.

[99] O. H. Hald, On discrete and numerical inverse Sturm-Liouville problems, Ph. D.

Thesis, New York University, 1972.

[100] O. H. Hald, Inverse eigenvalue problems for Jacobi matrices, Linear Alg. Appl., 14(1976), 63-85.

[101] L. A. Hageman and D. M. Young, Applied Iterative Methods, Academic Press, New York, 1981.

[102] S. P. Han, A successive projection method, Math. Programming, 40(1988), 1-14.

[103] U. Helmke and J. B. Moore, Optimization and Dynamical Systems, Springer- Verlag, London, 1994.

[104] Hershkowits, D., Existence of matrices with prescribed eigenvalues and entries, Linear and Multilinear Algebra, 14(1983), 315-342.

[105] N. J. Higham, Matrix nearest problems and applications, Proceedings of the IMA Conference on Applications of Matrix Theory, S. Barnett and M. J. C. Gover, eds., Oxford University Press, 1989, to appear.

[106] R. D. Hill, R. G. Bates and S. R. Waters, On centrohermitian matrices, SIAM J.

Matrix Anal. Appl., 11(1990), 128-133.

[107] M. Hirsch and S. Smale, On algorithms for solving f(x) = 0, Comm. Pure Appl.

Math., 32(1979), 281-312.

[108] H. Hochstadt, On some inverse problems in matrix theory, Arch. Math., 18(1967),201-207.

[109] H. Hochstadt, On construction of a Jacobi matrix from spectral data, Linear Alg.

Appl., 8(1974), 435-446.

[110] D. Hershkowits, Existence of matrices with prescribed eigenvalues and entries, Linear and Multilinear Algebra, 14(1983), 315-342.

[111] H. Hochstadt, On the construction of a Jacobi matrix from mixed given data, Linear Alg. Appl., 28(1979), 113-115.

[112] A. J. Hoffman and H. Wielandt, The variation of the spectrum of a normal matrix, Duke Math. J., 20(1953), pp. 37-39.

[113] R. A. Horn and C. A. Johnson, Matrix Analysis, Cambridge University Press, Cambridge, 1985.

[114] P. Horst, Factor Analysis of Data Matrices, Holt, Rinehart and Winston, New York, 1965.

[115] F. I. Karpelevič, On the characteristic roots of matrices with nonnegative elements, Izv. Akad. Nauk SSSR Ser. Mat. 15(1951), 361-383 (in Russian).

[116] H. B. Keller, Global homotopies and Newton methods, in Recent Advances in Numerical Analysis, C. de Boor and G. H. Golub, eds., Academic Press, New York, 1978, 73-94.

[117] P. E. Kloeden and J. Lorenz, A note on multistep methods and attracting sets of dynamical systems, Numer. Math., 56(1990), 667-673

[118] P. Lancaster, On eigenvalues of matrices dependent on a parameter, Numer. Math., 6, pp. 377-387.

[119] H. J. Landau, The inverse eigenvalue problem for real symmetric Toeplitz matrices, J. AMS, 7(1994), 749-767.

[120] D. P. Laurie, A numerical approach to the inverse Toeplitz eigenproblem, SIAM J.

Sci. Stat. Comput., 9(1988), 401-405.

[121] C. L. Lawson and R. J. Hanson, Solving Least Squares Problems, Prentice-Hall, New Jersey, 1974.

[122] T. Y. Li, On Chow, Mallet-Paret and Yorke, homotopy for solving systems of polynomials, Bulletin Inst. Math. Acad. Sinica, 11(1983), 433-437.

[123] T. Y. Li and N. H. Rhee, Homotopy algorithms for symmetric eigenvalue problems, Numer. Math., 55(1989), 265-280.

[124] T. Y. Li, H. Z. Sun and X. H. Sun, Parallel homotopy algorithm for symmetric tridiagonal eigenvalue problem, SIAM J. Sci. Stat. Comput., 12(1991), 469-487.

[125] N. G. Lloyd, Degree Theory, Cambridge University Press, New York, 1978. [126] R. Loewy and D. London, A note on an inverse problems for nonnegative matrices, Linear and Multilinear Algebra, 6(1978), 83-90.

[127] D. G. Luenberger, Introduction to Linear and Nonlinear Programming, Addison- Wesley, Reading, Mass., 1984.

[128] A. W. Marshall and I Olkin , Inequalities: Theory of Majorization and Its Applications, Academic Press, New York, 1979.

[129] V. Mehrmann and W. Rath, Numerical methods for the computation of analytic singular value decompositions, preprint, 1993.

[130] J. Milnor, Singular Points of Complex Hyper surfaces, Annals of Mathematics Studies 61, Princeton University Press, Princeton, New Jersey, 1968.

[131] H. Minc, Nonnegative matrices, Wiley, New York, 1988.

[132] L. Mirsky, Matrices with prescribed characteristic roots and diagonal elements, J.

London Math. Soc., 33(1958), 14-21.

[133] J. B. Moore, R. E. Mahony and U. Helmke, Numerical gradient algorithms for eigenvalue and singular value decomposition, SIAM J. Matrix Anal. Appl., 15(1994), 881-902.

[134] A. P. Morgan, A homotopy for solving polynomial systems, Applied Math. Comput., 18(1986), 87-92.

[135] J. Moser, Finitely many mass points on the line under the influence of an exponential potential — An integrable dynamical system, in Dynamical Systems Theory and Application, ed. J. Moser, Springer- Verlag, New York, 1975.

[136] M. Muller, An inverse eigenvalue problem computing B-stable Runge-Kutta methods having real poles, BIT, 32(1992), 676.

[137] Y. Nakamura, Group actions on linear predictors for nonstationary processes, IMA J. Math. Control Inform., 5(1988), 69-75.

[138] J. Nocedal and M. L. Overton, Numerical methods for solving inverse eigenvalue problems, Lect. Notes Math., 1005(1983), 212-226.

[139] G. N. Oliveira, Nonnegative matrices with prescribed spectrum, Linear Algebra Appl., 54(1983), 117-121.

[140] J. Paine, A numerical method for the inverse Sturm-Liouville problem, SIAM J.

Sci. Stat. Comput., 5(1984), 149-156.

[141] B. N. Parlett, The Symmetric Eigenvalue Problem, Prentice-Hall, Englewood Cliffs, N.J., 1980.

[142] J. D. Pryce, Numerical Solution of Sturm-Liouville Problems, Oxford University Press, New York, 1993.

[143] R. L. Parker and K. A. Whaler, Numerical methods for establishing solutions to the inverse problem of electromagnetic induction, J. Geophys. Research, 86(1981), 9574-9584.

[144] M. Renardy and R. C. Rogers, An Introduction to Partial Differential Equations, Springer-Verlag, New York, 1993.

[145] A. Ruhe, Closest normal matrix finally found!, BIT, 27(1987), 585-595.

[146] J. M. Sanz-Serna, Runge-Kutta schemes for Hamiltonian systems, BIT, 28(1988), 877-883.

[147] J. M. Sanz-Serna and M. P. Calvo, Numerical Hamiltonian Problems, Chapman and Hall, London, 1994.

[148] L. F. Shampine and M. K. Gordon, Computer Solution of Ordinary Differential Equations: The Initial Value Problem, Freeman, San Francisco, 1975.

[149] L. F. Shampine and M. W. Reichelt, The MATLAB ODE suite, preprint, 1995.

[150] A. Shapiro, On the unsolvability of inverse eigenvalue problems almost everywhere, Linear Alg. Appl., 49(1983), 27-41.

[151] S. Smale, A convergent process of price adjustment and global Newton methods, J. Math. Econ., 3(1976), 107-120.

[152] S. T. Smith, Geometric Optimization Methods for Adaptive Filtering, Ph.D. Thesis, Harvarad University, 1993.

[153] G. W. Soules, Constructing symmetric nonnegative matrices, Linear and Multilinear Algebra, 13(1983), 241-251.

[154] G. W. Stewart, The efficient generation of random orthogonal matrices with an application to condition estimators, SIAM Journal on Numerical Analysis, 17(1980), 403–404.

[155] G. W. Stewart and J. Sun, Matrix Perturbation Theory, Academic Press, Boston, 1990.

[156] J. Sun, Eigenvalues and eigenvectors of a matrix dependent on several parameters, J. Comput. Math., 3(1985), 351-364.

[157] W. W. Symes, The QR algorithm and scattering for the finite nonperiodic Toda lattice, Physica, 4D(1982), 275-280.

[158] J. F. Traub, Iterative Methods for the Solution of Equations, Prentice-Hall, Englewood Cliffs, N.J., 1964.

[159] W. F. Trench, Numerical solution of the eigenvalue problem for Hermitian Toeplitz matrices, SIAM J. Matrix Anal. Appl., 10(1989), 135-146.

[160] A. van der Sluis, Condition numbers and equilibration matrices, Numer. Math., 14(1969), 14-23.

[161] H. J. Wacker, Continuation Methods, Academic, New York, 1978.

[162] J. Y. Wang and B. S Garbow, A numerical method for solving inverse real symmetric eigenvalue problem, SIAM J. Sci. Stat. Comput., 4(1983), 45-51.

[163] D. S. Watkins, Isospectral Flows, SIAM Review, 26(1984), 379-391.

[164] L. T. Watson, S. C. Billups and A. P. Morgan, Algorithm 652: HOMPACK: A suite of codes for globally convergent homotopy algorithms, ACM Trans. Math. Software, 13(1987), 281-310.

[165] W. Wilhelmi, Eig Algorithmus zur Losung eines inversen Eigenwertproblems, ZAMM., 54(1974), 53-55.

[166] J. H. Wilkinson, The Algebraic Eigenvalue Problem, Clarendon Press, Oxford, 1965.

[167] S. S. Wilks, Mathematical Statistics, John Wiley and Sons, New York, 1962. [168] K. Wright, Differential equations for the analytic singular value decomposition of a matrix, Numer. Math., 3(1992), 283-295.

[169] Z. Y. Zhang, On the perturbation of the eigenvalues of a nondefective matrix, Math. Numer. Sinica, 8(1986), 106-108.